Author’s response to reviews

Title: BARRIERS AND ENABLERS FOR THE DEVELOPMENT AND IMPLEMENTATION OF ALLIED HEALTH CLINICAL PRACTICE GUIDELINES IN SOUTH AFRICAN PRIMARY HEALTH CARE SETTINGS: A QUALITATIVE STUDY

Authors:

Janine Margarita Dizon (janine.dizon@unisa.edu.au)
Karen Grimmer (Karen.grimmer@unisa.edu.au)
Quinette Louw (qalouw@sun.ac.za)
Shingai Machingaidze (shingai.machingaidze@mrc.ac.za)
Hameedah Parker (hparker@sun.ac.za)
Heath Pillen (heath.pillen@unisa.edu.au)

Version: 2 Date: 11 Aug 2017

Author’s response to reviews:

Dear Editors,

We made major revisions to the paper we are resubmitting and was not possible to do just track changes. However, we have highlighted the changes in red font in the revised manuscript document we are uploading for easier tracking of amendments and additions to the manuscript.

Thank you.

Janine and co-authors